Author's response to reviews

Title: Adjuvant breast cancer chemotherapy during late-trimester pregnancy: not quite a standard of care

Authors:

Richard J Epstein (repstein@hku.hk)

Version: 4 Date: 29 May 2007

Author's response to reviews:

All of the formatting revisions have been made, with the exception of the subsections stated in the revision letter for the Abstract and Manuscript. These state that the subsections should be: Background, Discussion, Summary. However, it is clearly stated in the Instructions to Authors that the subsections should be Background, Methods, Results/Discussion, Conclusions. Hence, this has been left unchanged.